Author's response to reviews

Title: Post-Deployment Health Care Needs in Small Community-Based Primary Care Clinics

Authors:

Polly H Noel (noelp@uthscsa.edu)
John E Zeber (John.Zeber@va.gov)
Mary J Pugh (pughm@uthscsa.edu)
Erin P Finley (finleye@uthscsa.edu)
Michael L Parchman (parchman@uthscsa.edu)

Version: 3 Date: 20 July 2011

Author's response to reviews:

July 20, 2011

Editors
BioMed Central – BMC Family Practice
Re: submission of reformatted manuscript #6908626365026469

Dear BMC Family Practice Editors:

Thank you for accepting our manuscript: A Pilot Survey of Post-Deployment Health Care Needs in Small Community-Based Primary Care Clinics for publication in BMC Family Practice. We have made the requested revisions, as outlined below:

Major revisions (we require the author to make these changes)

Title: please ensure that the title is identical in the manuscript and the submission system
The titles in the manuscript and submission system are now identical.

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs)
The author list has been revised to reflect the preferred style and entry into the submission system.

Abstract: This should be identical in the manuscript and the submission system
The abstracts in the manuscript and submission system are now identical.

Competing interests: Please include a `Competing interests¿ section between the Conclusions and Authors¿ contributions. If there are none to declare, please write `The authors declare that they have no competing interests¿.
Please see here for a full list of questions to consider when writing your competing interests statement:
http://www.biomedcentral.com/bmcmed/ifora/?txt_jou_id=2999&txt_mst_id=50733#interests

The competing interests section has been included.

Authors' contributions: Please use the following format as an example, including using initials to refer to each author's contribution: AB carried out the molecular genetic studies, participated in the sequence alignment and drafted the manuscript. JY carried out the immunoassays. MT participated in the sequence alignment. ES participated in the design of the study and performed the statistical analysis. FG conceived of the study, and participated in its design and coordination. All authors read and approved the final manuscript. This section should come after the competing interests and before the acknowledgements.

Author contributions have been added.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Header: Please remove this from the manuscript.

The header has been removed.

Title page: Please remove all the data about the document under the title (word counts etc). Please also remove all titles from the author list. Please move guarantor information to the acknowledgements and change the word "contact" to "corresponding author".

Data about the document under the title have been deleted. We assume the request to remove “titles” from the author list refers to the academic degrees. If so, these have been removed. Guarantor information has been moved to the acknowledgements & the word contact has been changed to corresponding author.

Affiliations: Please remember to include the country in the affiliation details.

Country has been added to affiliations.

Tracking: Please remove all tracking from the manuscript.

All tracking has been removed.

Structure: Please change the heading "Introduction" to "Background". Please also include a conclusions section after the discussion.

The “Introduction” heading has been changed to “Background.”

Tables: Please remove all vertical lines from tables.

The vertical lines have been removed from Table 1.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the
final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

The manuscript has been proofed for typographical errors.

Thank you again for acceptance of our manuscript. We look forward to its publication.

Sincerely yours,

Polly Hitchcock Noël, PhD (Corresponding Author)
Associate Director, VERDICT, South Texas Veterans Health Care System
Associate Professor, Dept. of Medicine, UT Health Science Center at San Antonio
7400 Merton Minter Blvd, San Antonio, Texas 78229
210-617-5314 (phone)
210-567-4423 (fax)
NOELP@UTHSCSA.EDU